Author's response to reviews

Title: Investigating a TELEmedicine solution to improve MEDication adherence in chronic Heart Failure (TELEMED-HF): study protocol of a randomized controlled trial

Authors:

Dionne Kessing (D.E.F.Kessing@uvt.nl)
Johan Denollet (J.Denollet@uvt.nl)
Jos Widdershoven (jwiddershoven@tsz.nl)
Nina Kupper (H.M.Kupper@uvt.nl)

Version: 5 Date: 13 October 2011

Author's response to reviews: see over
Dear Editors-in-Chief of Trials,

Thank you and the reviewer for your willingness to accept our manuscript number 1765144331591315.

The following changes requested for were made:

1. All authors email addresses on the title page of the manuscript are identical to those in the online submission system.
2. We inserted a Trials Status section after the Discussion section including details about the current status of the trial at the time of being submitted (see page 16 of the revised manuscript).
3. We checked our manuscript carefully for any typographical errors and to make our final revisions.

The manuscript now conforms to your journal style and we look forward to the final publication in Trials.

Yours sincerely,

Dionne Kessing and Nina Kupper on behalf of the authors

Corresponding author:
Nina Kupper, PhD, CoRPS – Center of Research on Psychology in Somatic diseases, Department of Medical Psychology & Neuropsychology, Tilburg University
PO Box 90153, 5000 LE Tilburg, the Netherlands
Tel: +31 13 466 2956 Fax: +31 13 466 2067
Email: H.M.Kupper@uvt.nl
Website: http://www.tilburguniversity.edu/corps